Author's response to reviews

Title: Validation of the Work Observation Method By Activity Timing (WOMBAT) method of conducting time-motion observations in critical care settings: an observational study

Authors:

Mark A Ballermann (mark.ballermann@ualberta.ca)
Nicola T Shaw (nicola.shaw@algomau.ca)
Damon C Mayes (damon.mayes@albertahealthservices.ca)
RT Noel Gibney (ngibney@ualberta.ca)
Johanna I Westbrook (j.westbrook@unsw.edu.au)

Version: 4 Date: 20 April 2011

Author's response to reviews:

Dear editors,

These modifications are in response to the requests from the editorial production team.

I have revised the figure titles to make them more concise.
I have removed the 'Objective' heading from the Abstract.
All highlighting and tracking has been removed from the manuscript.
The visible vertical lines have been removed from the tables.
The repeated column headings have been removed.
Roman numbers references to the tables have been removed.
The figures have been cropped to remove all extra white space.

Thank you very much for your time,

Dr. Mark Ballermann